Comments:

Although this is a retrospective work, it contributed a lot in ordering and simplifying the classification of injuries and favoring conscious decision-making.

Nice suggestions about approach to displastic polypoid lesions, thank you

Reply:

Dear reviewers and editor,

We have done careful reading and language polishing. We also edited the form of tables as required. Thank you very much.

Sincerely,

Yuan Li, Hanlin L. Wang